Author's response to reviews

Title: Splenic rupture as the presenting manifestation of primary splenic angiosarcoma: a case report

Authors:

Andreas Manouras (amanouras@hippocratio.gr)
Panagiotis Giannopoulos (giannopoulos@hotmail.com)
Levon Toufektzian (tlevon@gmail.com)
Haridimos Markogiannakis (markogiannakis@easy.com)
Emmanuel E Lagoudianakis (emlag@med.uoa.gr)
Artemisia Papadima (redemlag@yahoo.gr)
Dimitrios Papanikolaou (dpapanik81@yahoo.gr)
Konstantinos Filis (kfilis@hotmail.com)
Panagiotis Kekis (pkekis@hotmail.com)

Version: 2 Date: 18 January 2008

Author's response to reviews:

UNIVERSITY OF ATHENS
ATHENS MEDICAL SCHOOL
1st Department of Propaedeutic Surgery
January 18, 2008

Editorial Office,
Journal of Medical Case Reports

Dear Sir/Madam,

Please find attached the revised paper of ours entitled "Splenic rupture as the presenting manifestation of primary splenic angiosarcoma: a case report" for consideration for publication in Journal of Medical Case Reports.

Changes in the manuscript have been made according to your suggestions and the Journal's Instructions for Authors.

In particular:
1) The title of the paper has been changed so as to end with the following: "a case report." (Splenic rupture as the presenting manifestation of primary splenic angiosarcoma: a case report).
2) Corresponding Author has been marked with a *.
3) The keywords section has been removed.
4) The word "Background" has been corrected to "Introduction".
5) Our Consent section includes the following: "Written informed consent was obtained from the patient for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."

Yours sincerely,

Dr. Haridimos Markogiannakis, MD